Author's response to reviews

Title: Alcohol consumption and binge drinking in adolescents: comparison of different migration backgrounds and rural vs. urban residence - a representative study

Authors:

Carolin Donath (carolin.donath@uk-erlangen.de)
Elmar Gräßel (elmar.graessel@uk-erlangen.de)
Dirk Baier (baier@kfn.uni-hannover.de)
Christian Pfeiffer (kfn@kfn.uni-hannover.de)
Deniz Karagülle (karaquelle.deniz@mh-hannover.de)
Stefan Bleich (bleich.stefan@mh-hannover.de)
Thomas Hillemacher (hillemacher.thomas@mh-hannover.de)

Version: 3 Date: 14 December 2010

Author's response to reviews:

The sentence with the independent scientific consulting committee was omitted. Aims were included in the background section of the abstract.